
Site work today
v01: dropped EC100 box, reprogrammed to 20sps/10Hz filter, reinstalled box

v07: had to pio to get Li7500 running again.  Upper DSM probably needs ground strap.  Li7500 box probably needs ground strap.

v06: added ground strap to DSM. Li7500 box probably needs ground strap, though it was still running from this morning's reset.

tnw05: reseated Tsoil connectors multiple times.  Still reports 0.001 for all probes.  Will replace (stealing from tse05, that probably won't be erected).
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